Author's response to reviews

Title: Information from physicians and retention of information by patients - Obstacles to the awareness of patients of progressing disease when life is near to an end.

Authors:

Lena Hoff (Lena.Hoff@med.lu.se)
Goran Hermeren (Goran.Hermeren@med.lu.se)

Version: 5 Date: 27 December 2007

Author's response to reviews:

A few words have been corrected and the Additional material is located now to page 19.